# ORIGINAL RESEARCH

# Repeated Measurement of FIB-4 to Predict Long-Term Risk of HCC Development Up to 10 Years After SVR

Yanzheng Zou<sup>1,\*</sup>, Ming Yue<sup>2,\*</sup>, Linna Jia<sup>1</sup>, Yidi Wang<sup>1</sup>, Hongbo Chen<sup>3</sup>, Yifan Wang<sup>3</sup>, Meiling Zhang<sup>3</sup>, Yue Feng<sup>4</sup>, Rongbin Yu<sup>1</sup>, Sheng Yang<sup>5</sup>, Peng Huang<sup>1</sup>

<sup>1</sup>Department of Epidemiology, Center for Global Health, School of Public Health, Nanjing Medical University, Nanjing, People's Republic of China; <sup>2</sup>Department of Infectious Diseases, The First Affiliated Hospital of Nanjing Medical University, Nanjing, People's Republic of China; <sup>3</sup>Department of Infectious Disease, Jurong Hospital Affiliated to Jiangsu University, Jurong, People's Republic of China; <sup>4</sup>Faculty of Life Science and Technology, Kunming University of Science and Technology, Yunnan, People's Republic of China; <sup>5</sup>Department of Biostatistics, Center for Global Health, School of Public Health, Nanjing Medical University, Nanjing, People's Republic of China

\*These authors contributed equally to this work

Correspondence: Peng Huang, Department of Epidemiology, Center for Global Health, School of Public Health, Nanjing Medical University, Nanjing, People's Republic of China, Email huangpeng@njmu.edu.cn; Sheng Yang, Department of Biostatistics, Center for Global Health, School of Public Health, Nanjing Medical University, Nanjing, People's Republic of China, Email yangsheng@njmu.edu.cn

**Purpose:** It is unclear whether and how the long-term risk of hepatocellular carcinoma (HCC) will change in hepatitis C virus (HCV) infected patients who have reached sustained virologic response (SVR) with direct-acting antivirals (DAA). In this study, we assessed the long-term risk of HCC up to 10 years after SVR using fibrosis 4 score (FIB-4) and its dynamic changes.

**Patients and Methods:** A total of 701 DAA-treated patients who achieved SVR between January 2012 to October 2020 were enrolled in the study. The FIB-4 score of each patient was measured at the date of SVR and each follow-up visit annually. Patients were followed until December 31, 2021, with the longest follow-up time being 9.82 years.

**Results:** Following SVR, 27 cases of HCC were observed. The annual incidence rate of HCC remained stable with no obvious downward trend. Patients with a FIB-4 >3.25 at baseline or anytime during follow-up were at a higher risk of developing HCC than those whose FIB-4 remained below 3.25. Patients with cirrhosis and patients with no cirrhosis but a FIB-4 >3.25 were at higher risk of developing HCC than patients with no cirrhosis and a FIB-4  $\leq$ 3.25.

**Conclusion:** FIB-4 >3.25 measured at SVR or any time post-SVR was associated with HCC risks. The repeated measurement of FIB-4 revealed a better predictive ability of HCC risks than the simple measurement of FIB-4 at baseline. The additional stratification of patients by combining FIB-4 and cirrhosis leads to more accurately identifying high-risk patients. Surveillance of HCC is recommended for virologically cured patients with a FIB-4 >3.25 at SVR or anytime afterward and patients diagnosed with cirrhosis. **Keywords:** chronic hepatitis C, sustained virologic response, direct-acting antivirals, serum biomarker, hepatocellular carcinoma

# Plain Language Summary

In this study, we attempted to assess the long-term risk of hepatocellular carcinoma using the fibrosis 4 score (FIB-4) and its dynamic changes after the eradication of the hepatitis C virus. Our findings show that patients with a FIB-4 >3.25 at baseline or anytime during follow-up were at a higher risk of developing HCC. Patients with cirrhosis and patients with no cirrhosis but a FIB-4 >3.25 were at higher risk of developing HCC as well.

## Introduction

Hepatitis C virus (HCV) infection has been a severe public health problem for more than 30 years, with an estimated 71 million chronically infected worldwide.<sup>1</sup> Treatments with either interferon (IFN) or with direct-acting antivirals

Journal of Hepatocellular Carcinoma downloaded from https://www.dovepress.com/ For personal use only.



(DAA) offer remarkable rates of HCV virus eradication,<sup>2,3</sup> especially with DAA. The new DAA-based regimen has led to a sustained virologic response (SVR) rate of over 90% in patients with HCV infection.<sup>4</sup> With the widespread use of DAA treatment, it is foreseeable that most HCV patients will achieve SVR after their antiviral treatments.

One of the most severe outcomes of HCV infection is the development of hepatocellular carcinoma (HCC). HCC, the most common primary liver malignancy, resulted in approximately 830,000 deaths in 2020 alone.<sup>5</sup> Several novel medical options for HCC treatment have emerged in recent years, including immune checkpoint inhibitors (ICIs),<sup>6,7</sup> regorafenib (REG),<sup>8</sup> and metronomic capecitabine.<sup>9</sup> Despite these new treatment options, HCC continues to be a major threat to the survival of HCV patients.

Multiple studies have shown mitigation of liver fibrosis<sup>10,11</sup> and reduced risk of HCC<sup>12,13</sup> after patients reached SVR with DAA. However, the residual risk of HCC persists years after HCV eradication.<sup>14–16</sup> Therefore, surveillance for HCC is still needed even after patients achieve SVR. Several baseline characteristics, including older age,<sup>15</sup> male gender,<sup>17</sup> presence of cirrhosis,<sup>15,16</sup> excessive alcohol use,<sup>18,19</sup> and diabetes<sup>19</sup> were reported to be related to the risk of HCC development and can be used to narrow down patients in need of surveillance.

However, the majority only take into account the short-term risk of HCC. It is not entirely clear whether the risk of HCC will change over time in virologically cured patients. Fibrosis 4 (FIB-4) score is a non-invasive serum biomarker initially designed for the evaluation of fibrosis.<sup>20,21</sup> Composed of age, platelet counts, aspartate aminotransferase (AST), and alanine aminotransferase (ALT) levels, the FIB-4 score is easy to measure and calculate. In addition, it has been reported to be associated with the degree of liver fibrosis and the risk of HCC.<sup>22–24</sup> With the repeated measurement of FIB-4, it is possible to observe HCC risk changes over time and thus identify patients needing surveillance at any time after SVR.

The current study aimed to examine the association between FIB-4 score and its dynamic changes with the risk of HCC in virologically cured patients with no cirrhosis and compensated cirrhosis. We also determined whether the repeated measurement of FIB-4 could provide a good prediction of HCC and identified high-risk patients who require surveillance post-SVR.

#### **Material and Methods**

#### Patients and Follow-Up

A total of 1042 patients with chronic HCV infection from the Chronic Hepatitis C Research Program of Jiangsu (CHCRPJ) received DAA treatment from January 2012 to October 2020 in Jurong people's hospital, China. SVR was defined as a serum HCV RNA viral load below the lower detection limit at least 12 weeks after treatment. Of these patients, 796 achieved SVR. Patients diagnosed with HCC or hepatic decompensation (defined as the occurrence of ascites, encephalopathy, variceal bleeding, or hepatorenal syndrome) prior to SVR and patients who lack a FIB-4 score at baseline were excluded. Eventually, 701 patients were included in the study. The flowchart of patient selection is shown in Figure 1.

The index date of the study was defined as the date of SVR. Patients were followed until the study outcome, death or 31/12/2021, whichever came first. The study outcome was new cases of HCC after the index date. Information on HCC occurrence before and after treatment was obtained from hospital inpatient and outpatient diagnoses. HCC was diagnosed according to the American Association for the Study of Liver Diseases (AASLD) guidelines.<sup>25</sup> Laboratory tests were carried out at the date of SVR and annually after that when patients returned for their follow-up visit.

Written informed consent was obtained from all patients who participated in the study. The study protocol conformed to the ethical guidelines of the Declaration of Helsinki, and the institutional ethics review committee of Nanjing Medical University approved the study.

#### **Baseline Characteristics**

Baseline characteristics at SVR, including age, gender, cirrhosis, diabetes, hepatitis B virus (HBV) co-infection, and alcohol use, were collected. The birthdate and gender of each patient were obtained from their identity profiles. Information about cirrhosis, HBV co-infection, and diabetes was retrieved from the inpatient and outpatient diagnoses before SVR. Cirrhosis was diagnosed based either on a transient elastography score >14 kPa, a liver biopsy showing



Figure I Flow chart of patient selection.

Abbreviations: SVR, sustained virological response; HCC, hepatocellular carcinoma; FIB-4, fibrosis 4 score.

Metavir F4, or clinical evidence. Information about alcohol use was collected through telephone inquiries. Excessive alcohol use was defined as >20 g/day in females and >30 g/day in males.

## Fibrosis-4 (FIB-4) Scores at Baseline and During Follow-Up

Baseline FIB-4 score was calculated using the laboratory test results done at the same time as SVR, according to the following formula:  $(Age \times AST)/$  (Platelet count× $\sqrt{ALT}$ ).<sup>26</sup> FIB-4 score during follow-up was calculated based on the laboratory test results performed at the annual visit of each patient. The cutoff value for FIB-4 level was set at 3.25, which was previously established as the threshold of advanced fibrosis<sup>21</sup> and was shown to be associated with HCC risks.<sup>23,27</sup>

The changes of FIB-4 over time were modeled in two ways. First, we analyzed the FIB-4 score of patients at baseline and the last follow-up, then we categorized patients into four groups. For patients who developed HCC during follow-up, the last follow-up was defined as the time of HCC occurrence.

- (a) Patients with a FIB-4  $\leq$  3.25 at baseline and last follow-up were classified in the unchanged low-risk group.
- (b) Patients who dropped from FIB-4 >3.25 at baseline to FIB-4 ≦3.25 at the last follow-up were classified in the declined group.
- (c) Patients with a FIB-4 ≤3.25 at baseline and a FIB-4 >3.25 at the last follow-up were categorized as the increased group.
- (d) Patients with a FIB-4 >3.25 at baseline and last follow-up were categorized as the unchanged high-risk group.

Second, we attempted to examine the impact of changes in FIB-4 score throughout the entire study period. To achieve that, we analyzed FIB-4 level (categorized as >3.25 or  $\leq$ 3.25) as a time-dependent covariate. Each patient was analyzed under the low FIB-4 category when their FIB-4 score was below 3.25 and under the high FIB-4 category when their FIB-4 score was above 3.25.

## Statistical Analyses

Continuous variables were presented as mean (standard deviation), and categorical variables were presented as count (percentage). Continuous variables were compared using the Student's *t*-test, and categorical variables by the Chi-square test or the Fisher exact test when appropriate.

The annual incidence rate of HCC (per 1000 person-years [PY]) and the corresponding 95% confidence interval (95% CI) were calculated assuming a Poisson distribution. The annual incidence rate of HCC between subgroups and between years after SVR was compared using two-way ANOVA with Tukey post-hoc analysis. The cumulative incidence of HCC

was evaluated using the Kaplan-Meier method, and the differences between subgroups were compared with the log-rank test.

Univariate and multivariate Cox proportional hazards regression models were used to estimate the effects of potential factors on the risk of HCC occurrence. Variables found to be statistically significant in univariate analysis (P < 0.05) were included in the multivariate analysis. HCC risks were also calculated according to the change in FIB-4 score at the last follow-up and based on FIB-4 coded as a time-dependent covariate using Cox regression models. Patients not developing HCC were censored on 31/12/2021 or at the date of death.

Statistical significance was defined as P < 0.05. Data analysis was performed using R software, version 4.1.2 (R Foundation for Statistical Computing).<sup>28</sup>

## Results

#### Baseline Characteristics of Patients

The baseline characteristics of the study patients are shown in Table 1. Among the 701 patients, the majority (75.5%) were female. The mean age was 56.7 years old (SD = 8.3 years). 14.4% of the patients had cirrhosis at baseline. 24.5% of the patients had diabetes, 12.0% had excessive alcohol consumption, and 1.4% had HBV co-infection. A total of 273 patients (39.0%) had a FIB-4 score above 3.25 at the initiation of treatment. Aside from cirrhosis status and FIB-4 level, there were no significant differences in baseline factors between patients who developed HCC and patients who did not.

The median follow-up time for the overall study population was 4.84 years, with the 25th and 75th percentiles of the follow-up time being 3.45 years and 6.65 years, respectively. The patient with the longest follow-up period was followed for up to 9.82 years.

## Incidence of HCC Following SVR

There were 27 cases of HCC following SVR. HCC incidence rate was 2.86 per 1000 PY during the first year, gradually increasing to 15.59 per 1000 PY in the third year and then decreasing afterward (Figure 2A). The cumulative incidence at 1, 2, 3, 4, and 5 years after SVR were 0.3%, 1.0%, 2.4%, 3.0%, and 3.3%, respectively.

We also plotted the annual incidence of HCC based on the presence or absence of baseline cirrhosis due to the possible difference between these two subgroups. The HCC incidence rate was stable in patients without cirrhosis, ranging from 1.67 per 1000 PY to 12.63 per 1000 PY. Patients with baseline cirrhosis had a noticeably higher annual incidence rate, with the highest incidence rate (33.98 per 1000PY) occurring three years after SVR (Figure 2B). There was also a slight rebound in the incidence rate in the fifth year after SVR (21.09/1000PY). In general, there was no significant downward trend in the annual incidence of HCC following SVR. The two-way ANOVA analysis discovered

Characteristics	All Patients (N=701)	Patients with HCC (N=27)	Patients Without HCC (N=674)	P value
Age, y, mean (SD)	56.7 (8.3)	56.6 (8.1)	56.7 (8.3)	0.931
Gender (%)				
Male	172 (24.5)	6 (22.2)	166 (24.6)	0.955
Female	529 (75.5)	21 (77.8)	508 (75.4)	
Cirrhosis (%)	101 (14.4)	8 (29.6)	93 (13.8)	0.044
HBV co-infection	10 (1.4)	I (3.7)	9 (1.3)	0.849
(%)				
Diabetes (%)	172 (24.5)	8 (29.6)	164 (24.3)	0.690
Excessive alcohol	84 (12.0)	2 (7.4)	82 (12.2)	0.657
use (%)				
FIB-4 (%)				
≦3.25	428 (61.1)	10 (37.0)	418 (62.0)	0.016
>3.25	273 (39.0)	17 (63.0)	256 (38.0)	

Table I Baseline Characteristics of Patients

Abbreviations: HBV, hepatitis B virus; FIB-4, fibrosis 4 score.



Figure 2 The annual incidence rate of HCC in virologically cured patients after SVR in the overall cohort (A) and the subgroups defined according to the presence of cirrhosis (B).

Abbreviations: SVR, sustained virological response; CI, confidence interval; FIB-4, fibrosis 4 score; HCC, hepatocellular carcinoma.

a substantial difference in the HCC incidence rate between the two subgroups (P = 0.006). No difference was found between years after SVR (all P > 0.05) in the Tukey post-hoc test.

#### Baseline Factors Associated with the Risk of HCC

Univariate and multivariate Cox regression models were used to evaluate baseline factors that might contribute to the development of HCC. Gender, excessive alcohol use, diabetes, cirrhosis, HBV co-infection, and FIB-4 score were included in the analysis. In the univariate analysis, cirrhosis (hazard ratio [HR] = 2.98; 95% CI = 1.29–6.86) and FIB-4 >3.25 (HR = 3.00; 95% CI = 1.37–6.56) were associated with a higher risk of HCC in patients achieving SVR. Significant variables (P < 0.05) in the univariate analysis were included in the multivariate analysis, where FIB-4 >3.25 (HR = 2.51; 95% CI = 1.10–5.72) was identified as an independent factor that contributed to the risk of HCC (Table 2) whereas cirrhosis was not.

Figure 3 depicts the cumulative incidence of HCC in patients who achieved SVR based on risk factors identified in the Cox regression analysis. The incidence of HCC was significantly higher in patients with a FIB-4 >3.25 at baseline than in those with a FIB-4  $\leq$ 3.25 (Figure 3A, *P* = 0.0044). Similarly, the incidence of HCC after SVR was higher in patients with cirrhosis (Figure 3B, *P* = 0.0085).

Characteristics	Univariate		Multivariate		
	HR (95% CI)	P value	HR (95% CI)	P value	
Gender, male vs female	1.13 (0.46,2.80)	0.791			
Excessive alcohol use, no vs yes	0.64 (0.15,2.69)	0.538	-	-	
Diabetes, no vs yes	1.45 (0.63,3.32)	0.377	-	-	
Cirrhosis, no vs yes	2.98 (1.29,6,86)	0.010	2.11 (0.88,5.08)	0.095	
HBV co-infection, no vs yes	3.04 (0.41,22.46)	0.275	_	-	
FIB-4, ≦3.25 vs >3.25	3.00 (1.37,6.56)	0.006	2.51 (1.10,5.72)	0.029	

Table 2 Baseline Factors Associated with HCC Risk in Patients Achieving SVR

Abbreviations: HCC, hepatocellular carcinoma; CI, confidence interval; HR, hazard ratio; SVR, sustained virologic response; HBV, hepatitis B virus; FIB-4, fibrosis 4 score.



Figure 3 Cumulative incidence curves of HCC in virologically cured patients after SVR according to different subgroups: (A) FIB-4 level; (B) cirrhosis. Abbreviations: SVR, sustained virological response; FIB-4, fibrosis 4 score; HCC, hepatocellular carcinoma.

#### HCC Risks Stratified by Baseline FIB-4 and Cirrhosis

In addition, we examined the association between baseline FIB-4 level together with cirrhosis and HCC risks. The incidence rate of HCC was the highest in patients with baseline cirrhosis and a FIB-4 >3.25 (18.63/1000PY). After adjustment for other baseline characteristics by the Cox regression model, patients with cirrhosis and a FIB-4 >3.25 (18.63/1000PY). After adjustment for other developing HCC than patients with no cirrhosis and a FIB-4  $\leq$ 3.25 (adjusted HR = 6.56; 95% CI = 2.13–20.19). Patients with baseline cirrhosis and a FIB-4  $\leq$ 3.25, were also at higher risk of HCC compared with patients with no cirrhosis and a FIB-4  $\leq$ 3.25 (adjusted HR = 5.50 and 3.19, respectively) (Table 3).

Figure 4 depicts the impact of FIB-4 level on the HCC-free survival of patients stratified by baseline cirrhosis. Among patients with no cirrhosis at baseline, the incidence of HCC was significantly higher in patients with a FIB-4  $\geq$ 3.25 than in patients with a FIB-4  $\leq$ 3.25 (Figure 4A, P = 0.012). No difference was observed in HCC incidence between patients with different FIB-4 levels among patients with cirrhosis (Figure 4B, P = 0.96).

#### Changes in FIB-4 Over Time and Its Effect on HCC Risks

We compared the FIB-4 scores of patients at baseline and the last follow-up visit. Of the 701 patients included in this study, the FIB-4 score at the last follow-up was available in 514 patients. In total, 56.4% of patients remained at low risk, and 18.9% remained at high risk at the last follow-up. The FIB-4 score decreased in 20.0% of patients and increased in only 4.7%. The incidence rate was highest in the increased group (29.60/1000PY), followed by the unchanged high-risk group (15.16/1000PY) and the declined group (12.52/1000PY). Patients who experienced an elevation in FIB-4 score had the highest risk of developing HCC (adjusted HR = 9.88; 95% CI = 2.56–38.14). Also, HCC risks remained high in patients with a baseline FIB-4 >3.25 regardless of whether the FIB-4 score had dropped at the last follow-up (adjusted HR = 3.97; 95% CI = 1.24-12.74) or not (adjusted HR = 4.99; 95% CI = 1.44-17.26) (Table 4).

Characteristics	Number of Patients (%)	Number Who Developed HCC (%)	Patient-Years	HCC Per 1000 Patient-Years	Crude HR (95% CI)	Adjusted HR (95% CI) <sup>†</sup>
No cirrhosis with FIB-4≦3.25	402 (67.0)	8 (2.0)	2274	3.52	I	I
No cirrhosis with FIB-4>3.25	198 (33.0)	11 (5.6)	1009	10.90	3.06 (1.23,7.61)	3.19 (1.27,7.99)
Cirrhosis with FIB-4≦3.25	26 (25.7)	2 (7.7)	128	15.63	4.40 (0.93,20.80)	5.50 (1.13,26.81)
Cirrhosis with FIB-4>3.25	75 (74.3)	6 (8.0)	322	18.63	5.12 (1.75,14.92)	6.56 (2.13,20.19)

Table 3 Association Between Baseline FIB-4 Level and HCC Risk Stratified by the Presence of Cirrhosis in Patients Achieving SVR

Note: <sup>†</sup>Adjusted for gender, excessive alcohol use, diabetes, and HBV co-infection.

Abbreviations: HCC, hepatocellular carcinoma; CI, confidence interval; HR, hazard ratio; SVR, sustained virologic response; FIB-4, fibrosis 4 score.



Figure 4 Cumulative incidence curves of HCC among virologically cured patients with different FIB-4 levels, stratified by cirrhosis: (**A**) cirrhosis; (**B**) without cirrhosis. Abbreviations: SVR, sustained virological response; FIB-4, fibrosis 4 score; HCC, hepatocellular carcinoma.

#### Association Between FIB-4 as a Time-Dependent Covariate and HCC Risks

We analyzed FIB-4 score (classified as >3.25 or  $\leq 3.25$ ) as a time-dependent covariate. The HCC incidence rate was 4.37 per 1000PY when the FIB-4 level of patients was low and 15.18 per 1000PY when the FIB-4 level of patients was high. During the entire follow-up period, FIB-4 >3.25 was associated with a 3.14-fold higher risk of patients developing HCC (adjusted HR = 3.14; 95% CI = 1.40–7.05).

When taking into account the presence of baseline cirrhosis, we discovered a surprisingly high incidence rate of HCC in patients with cirrhosis when their FIB-4 level was high (23.44/1000PY). The incidence rate was slightly higher in patients with cirrhosis and a FIB-4 score below 3.25 (10.26/1000PY) than in patients with no cirrhosis but a FIB-4 above 3.25 (12.55/1000PY). Among patients with cirrhosis, FIB-4 >3.25 during the follow-up period was associated with a 7.49-fold higher risk

Characteristics	Number of Patients (%)	Number Who Developed HCC (%)	Patient-Years	HCC Per 1000 Patient-Years	Crude HR (95% CI)	Adjusted HR (95% CI) <sup>†</sup>
Unchanged, low risk	290 (56.4)	5 (1.7)	1781	2.80	1	1
Declined	103 (20.0)	7 (6.80)	559	12.52	4.31 (1.37,13.60)	3.97 (1.24,12.74)
Increased	24 (4.7)	4 (16.7)	135	29.60	10.59 (2.84,39.46)	9.88 (2.56,38.14)
Unchanged, high risk	97 (18.9)	7 (7.2)	462	15.16	5.08 (1.60,16.10)	4.99 (1.44,17.26)

Table 4 Association Between Change in FIB-4 Level at Baseline and Last Follow-Up and HCC Risk in Patients Achieving SVR

Note: <sup>†</sup>Adjusted for gender, cirrhosis, excessive alcohol use, diabetes, and HBV co-infection.

Abbreviations: HCC, hepatocellular carcinoma; CI, confidence interval; HR, hazard ratio; SVR, sustained virologic response; FIB-4, fibrosis 4 score.

Table 5Association Between FIB-4 Level During Follow-Up and HCC Risk Stratified by the Presence of Cirrhosis in PatientsAchieving SVR

Characteristics	Patient-Years	HCC Per 1000 Patient-Years	Crude HR (95% CI)	Adjusted HR (95% CI) <sup><math>\dagger</math></sup>
No cirrhosis with FIB-4≦3.25	2566	3.90	I	I
No cirrhosis with FIB-4>3.25	717	12.55	3.37 (1.37,8.31)	3.29 (1.33,8.16)
Cirrhosis with FIB-4≦3.25	195	10.26	2.56 (0.56,11.72)	2.83 (0.58,13.72)
Cirrhosis with FIB-4>3.25	256	23.44	5.94 (2.14,16.53)	7.49 (2.56,21.95)

Note: <sup>†</sup>Adjusted for gender, excessive alcohol use, diabetes, and HBV co-infection.

Abbreviations: HCC, hepatocellular carcinoma; CI, confidence interval; HR, hazard ratio; SVR, sustained virological response; FIB-4, fibrosis 4 score.

of HCC (adjusted HR = 7.49; 95% CI = 2.56-21.95). In contrast, no increased risk of HCC development was found in cirrhosis patients when their FIB-4 scores were below 3.25 during follow-up (adjusted HR = 2.83; 95% CI = 0.58-13.72) (Table 5).

## Discussion

As a result of the wide use of DAA, almost all patients with HCV infection will experience SVR after antiviral treatment. Multiple studies have shown that SVR is associated with improved clinical outcomes, including the mitigation of fibrosis<sup>10,11</sup> and the reduced risk of HCC.<sup>12,29,30</sup> However, there is still a residual risk of HCC after SVR, so surveillance of patients is required. As described by Wang S et al<sup>31</sup> surveillance and early diagnosis are critical in improving the cure rate of cancer as they facilitate a timely and sufficient subsequent treatment. However, surveillance on all virologically cured patients is impractical. As a result, it is necessary to determine which of the patients are at high risk of developing HCC. In addition, the HCC risks may change over time in the long term as patients age and their fibrosis status changes, further complicating the situation. Several studies have evaluated pre-SVR factors contributing to the short-term risk of HCC in virologically cured patients, but few have addressed the long-term risk of HCC. Therefore, in the current study, we seek to use FIB-4, a simple serum fibrosis biomarker that can be repeatedly measured during follow-up, to evaluate the long-term risk of HCC and ultimately identify high-risk patients who require strict surveillance.

In this study, 701 HCV-infected patients treated with DAA were followed for a median of 4.84 years after SVR. The patient with the longest follow-up time was followed for up to 9.82 years. The study's first finding was that the risk of HCC persisted for years after SVR. After inspecting the annual incidence rate of HCC, we found no apparent decline in the residual HCC risks following SVR. The annual incidence rate in the overall cohort was below the cutoff value of 15 per 1000PY, as recommended by the AASLD.<sup>25</sup> Among patients with cirrhosis, however, the annual incidence rate of HCC was higher than the surveillance threshold most of the time, with a slight rebound in the fifth year after SVR. The two-way ANOVA test results revealed a significant difference in HCC incidence rate between patients with or without cirrhosis but no difference between years after SVR. The results above suggest that the risk of HCC after SVR did not reduce over time and that there was a significant difference in HCC risk between different subgroups of patients. The result is consistent with the study investigating HCC risk after HCV eradication in Veterans Health Administration (VHA) patients. The risk of HCC persisted 10 years after SVR and showed no indication of a downward trend in VHA patients.<sup>23</sup>

Secondly, the study found that FIB-4 >3.25 measured at baseline or any time after SVR was associated with HCC risks and that the repeated measurements of FIB-4 presented a better predictive ability of HCC risks than the simple measurement of FIB-4 at baseline. Univariate and multivariate Cox regression analysis showed that FIB-4 >3.25 at baseline was significantly associated with HCC development after HCV eradication. When analyzed as a time-dependent covariant, FIB-4 >3.25 showed a good predicting value of HCC as well. Patients faced a significantly higher risk of HCC when their FIB-4 was above 3.25 at any time during follow-up. The results suggest that FIB-4 measured at any time after treatment can be used to examine HCC risks in patients achieving SVR. Similar results were reported in previous studies as well. In a study evaluating the ability of blood fibrosis tests to predict liver-related events and death, FIB-4 showed good discriminative ability for predicting HCC (with a C-index of 0.884) and was an independent predictor of HCC in the adjusted Cox model.<sup>32</sup> Another study examining the long-time risk of HCC after HCV eradication found that the incidence of HCC was significantly higher in patients with a FIB-4  $\geq$  3.25.<sup>33</sup>

As expected, the FIB-4 level of most patients decreased or remained the same post-SVR. Only 4.7% of the patients experienced an increase in FIB-4 level from a value  $\leq 3.25$  to a value >3.25. The increase in FIB-4 level was associated with a 9.88-fold higher risk of HCC. A study focusing on HCC risk after HCV eradication found that an increase in FIB-4 was associated with HCC risk.<sup>23</sup> Though a rise in FIB-4 after SVR is relatively uncommon, the high incidence rate of HCC (29.6/1000PY) in these patients still requires attention. Repeated measurement of FIB-4 after treatment made it possible to identify this small group of patients at risk. In addition, the coefficient of FIB-4 level was larger in the time-dependent analysis than in the model composed of only baseline values, indicating that repeated measurement of FIB-4 can better predict HCC risks in SVR patients compared with measuring FIB-4 only at baseline.

Another finding of the current study was that compared to stratifying patients by cirrhosis status at baseline alone, the inclusion of FIB-4 considerably improved the ability to predict HCC risks in patients reaching SVR. When investigating

the risk of HCC occurrence, a number of studies simply stratified patients according to the presence of cirrhosis.<sup>14,34</sup> However, in the current study, baseline cirrhosis was only significantly associated with HCC risks in the univariate model, and HCC still occurred in patients without cirrhosis. Also, multiple studies have shown the changes in liver fibrosis status after patients achieved SVR.<sup>12,22</sup> It appears that the simple stratification based on cirrhosis is not sufficient for predicting HCC. Therefore, the current study combined baseline cirrhosis with FIB-4 and its dynamic changes to stratify virologically cured patients into distinct risk groups. In the baseline analysis, the incidence rate of HCC was the highest in patients with both cirrhosis and a FIB-4 >3.25 (18.63/1000PY). Cirrhotic patients and non-cirrhotic patients with a baseline FIB-4 above 3.25 were at a higher risk of developing HCC than non-cirrhotic patients with a baseline FIB-4 below 3.25. A large cohort study examining HCC risk 10 years after SVR also found that patients with cirrhosis or FIB-4 >3.25 at baseline presented a higher risk of developing HCC.<sup>23</sup> The time-dependent analysis delivered similar results. The incidence rate of HCC was very high when cirrhotic patients had a FIB-4 above 3.25 (23.44/1000PY). Also, when non-cirrhotic patients had a FIB-4 score higher than 3.25, their incidence rate of HCC (12.55/1000PY) was slightly higher than that of cirrhotic patients (10.26/1000PY) when the FIB-4 was below 3.25. These results all suggest that patients faced a significantly higher risk of HCC when their FIB-4 was above 3.25 at any time after HCV eradication, regardless of the presence of baseline cirrhosis. The additional segmentation of patients by combining FIB-4 and cirrhosis leads to more accurate identification of high-risk patients after the eradication of HCV.

Finally, the current study identified some patients at high risk of HCC development. Virologically cured patients with a FIB-4 level above 3.25 at SVR or any time afterwards are at a higher risk of developing HCC and should therefore be offered surveillance. Further study will be needed to examine whether the HCC risks will decrease in virologically cured patients experiencing a drop in FIB-4 level. In addition, patients with cirrhosis should continue to be surveilled after SVR, along with patients with no cirrhosis but a FIB-4 >3.25. Surprisingly, no increased risk was found in patients with baseline cirrhosis and a FIB-4  $\leq$ 3.25 in the time-dependent analysis. This may be partly explained by the small number of patients with both cirrhosis and a FIB-4  $\leq$ 3.25.

Compared with previous studies, the current study holds the following strengths. All patients included in the study were treated with DAA and achieved SVR. The follow-up period was also relatively long, with a median follow-up time of approximately 5 years and a maximum follow-up time of up to 10 years. Previous studies investigating the predictive value of FIB-4 were either done on IFN-treated patients<sup>33</sup> or did not have a sufficiently long follow-up duration of DAA-treated patients.<sup>23,27</sup> Our study demonstrated that the risk of HCC did not decline for a considerable time after DAA treatment and that the dynamic changes in FIB-4 levels were directly associated with HCC risk throughout the follow-up period. Furthermore, the current study combined the dynamic change of FIB-4 and cirrhosis status to divide patients into subgroups further, bringing attention to two particular groups of patients. (1) Patients whose FIB-4 was below 3.25 at the time of SVR but later increased to above 3.25 require strict surveillance as the HCC incidence rate in these patients was extremely high. This specific group of patients was often overlooked since some previous studies only evaluated FIB-4 measured at baseline.<sup>14,35</sup> (2) Patients with baseline cirrhosis but a FIB-4 ≦3.25 at or after SVR still need continuous surveillance despite their relatively low FIB-4 scores. Unlike some studies proposing that patients with a FIB-4  $\leq 3.25$  may be excluded from surveillance,<sup>27,33</sup> our results showed that there was still insufficient evidence to terminate surveillance for patients with low FIB-4 scores but other high-risk factors such as cirrhosis. The incidence rate of HCC in these patients remained relatively high, despite the timedependent analysis not identifying an increased risk. Further study and a larger population will be needed to investigate the long-term risk of HCC in this distinct group of patients.

The study has some limitations. First, some patients did not return for laboratory tests after SVR as time went by, and the association between FIB-4 change and HCC risks might be underrated or overrated. Second, information about alcohol use was collected through telephone inquiries rather than face-to-face questionnaires, which may have yielded information bias. Third, the current study was a single-center study, and the generalizability of our results to a larger HCV population may be limited. Finally, the study cohort is all Asian and consists mainly of females. It is necessary to conduct validation studies in other cohorts.

The role of FIB-4 in HCC prediction has been thoroughly described in the current study and previous literature. Future studies could put more emphasis on the following aspects. A larger population and a more extended follow-up period will be needed to determine the utility of FIB-4 or other non-invasive biomarkers in identifying cirrhotic patients at low enough risk to

reduce surveillance after SVR. In addition to single non-invasive biomarkers, new risk prediction models constructed based on longitudinal data collected during follow-up, capable of making a more personalized and dynamic prediction of HCC occurrence, will be instrumental in guiding the clinical decisions for HCV patients achieving SVR. Lastly, it is crucial that we concentrate on the urgent need to develop novel methods for HCC treatment. New techniques that tackle cancer in a different approach, such as the OncoCiDia strategy,<sup>31</sup> may hold a curative potential for HCC.

## Conclusion

In conclusion, the risk of HCC persists years after SVR. The repeated measurement of FIB-4 presents a better predictive ability of HCC risks than the simple measurement of baseline FIB-4. Patients with a FIB-4 level above 3.25 at SVR or any time after SVR are at a higher risk of developing HCC and should therefore be offered surveillance. By stratifying patients reaching SVR into different risk groups using a combination of FIB-4 and cirrhosis, high-risk patients can be identified more accurately. Patients with cirrhosis should continue to be surveilled after SVR, along with patients with no cirrhosis but a FIB-4 >3.25.

## **Abbreviations**

FIB-4, fibrosis 4 score; HCV, hepatitis C virus; SVR, sustained virologic response; IFN, interferon; DAA, direct-acting antivirals; HCC, hepatocellular carcinoma; ICIs, immune checkpoint inhibitors; REG, regorafenib; AST, aspartate aminotransferase; ALT, alanine aminotransferase; HBV, hepatitis B virus; PY, person-years; CI, confidence interval; HR, hazard ratio; AASLD, American Association for the Study of Liver Diseases; VHA, Veterans Health Administration.

## Acknowledgments

This study is a joint effort of many investigators and staff members, and their contribution is gratefully acknowledged. We especially thank all patients who participated in this study. This study was sponsored by the Natural Science Foundation of China (82273741), the Natural Science Foundation of Jiangsu Province (BK20190106), the Priority Academic Program Development of Jiangsu Higher Education Institutions (PAPD), and the Nanjing Important Science & Technology Specific Projects (2021-11005).

## Funding

This study was sponsored by the Natural Science Foundation of China (82273741), the Natural Science Foundation of Jiangsu Province (BK20190106), the Priority Academic Program Development of Jiangsu Higher Education Institutions (PAPD), and the Nanjing Important Science & Technology Specific Projects (2021-11005).

# Disclosure

The authors report no conflicts of interest in this work.

# References

- 1. European Association for the Study of the Liver. EASL recommendations on treatment of hepatitis C: final update of the series A. J Hepatol. 2020;73 (5):1170–1218. doi:10.1016/j.jhep.2020.08.018
- 2. Chang KC, Tung SY, Wei KL, et al. Real-world efficacy and safety of pangenotypic direct-acting antivirals against hepatitis C virus infection in Taiwan. *Sci Rep.* 2021;11(1):13543. doi:10.1038/s41598-021-93095-x
- 3. Fried MW, Shiffman ML, Reddy KR, et al. Peginterferon alfa-2a plus ribavirin for chronic hepatitis C virus infection. N Engl J Med. 2002;347(13):975-982.
- 4. Baumert TF, Berg T, Lim JK, Nelson DR. Status of direct-acting antiviral therapy for hepatitis C virus infection and remaining challenges. *Gastroenterology*. 2019;156(2):431–445. doi:10.1053/j.gastro.2018.10.024
- Deo SVS, Sharma J, Kumar S. GLOBOCAN 2020 report on global cancer burden: challenges and opportunities for surgical oncologists. Ann Surg Oncol. 2022;29(11):6497–6500. doi:10.1245/s10434-022-12151-6
- 6. Rizzo A, Ricci AD, Gadaleta-Caldarola G, Brandi G. First-line immune checkpoint inhibitor-based combinations in unresectable hepatocellular carcinoma: current management and future challenges. *Expert Rev Gastroenterol Hepatol*. 2021;15(11):1245–1251.
- 7. Rizzo A, Ricci AD, Di Federico A, et al. Predictive biomarkers for checkpoint inhibitor-based immunotherapy in hepatocellular carcinoma: where do we stand? *Front Oncol.* 2021;11:803133. doi:10.3389/fonc.2021.803133
- 8. Rizzo A, Nannini M, Novelli M, Dalia Ricci A, Scioscio VD, Pantaleo MA. Dose reduction and discontinuation of standard-dose regorafenib associated with adverse drug events in cancer patients: a systematic review and meta-analysis. *Ther Adv Med Oncol.* 2020;12:1758835920936932. doi:10.1177/1758835920936932

- De Lorenzo S, Tovoli F, Barbera MA, et al. Metronomic capecitabine vs. best supportive care in Child-Pugh B hepatocellular carcinoma: a proof of concept. Sci Rep. 2018;8(1):9997. doi:10.1038/s41598-018-28337-6
- 10. Rockey DC, Friedman SL. Fibrosis regression after eradication of hepatitis C virus: from bench to bedside. *Gastroenterology*. 2021;160(5). doi:10.1053/j.gastro.2020.09.065
- 11. Soliman H, Ziada D, Salama M, et al. Predictors for fibrosis regression in chronic HCV patients after the treatment with DAAS: results of a real-world cohort study. *Endocr Metab Immune Disord Drug Targets*. 2020;20(1):104-111.
- Ioannou GN, Feld JJ. What are the benefits of a sustained virologic response to direct-acting antiviral therapy for hepatitis C virus infection? Gastroenterology. 2019;156(2). doi:10.1053/j.gastro.2018.10.033
- Nahon P, Bourcier V, Layese R, et al. Eradication of hepatitis C virus infection in patients with cirrhosis reduces risk of liver and non-liver complications. *Gastroenterology*. 2017;152(1):142–156 e2. doi:10.1053/j.gastro.2016.09.009
- Kanwal F, Kramer J, Asch SM, Chayanupatkul M, Cao Y, El-Serag HB. Risk of hepatocellular cancer in HCV patients treated with direct-acting antiviral agents. *Gastroenterology*. 2017;153(4):996–1005.e1. doi:10.1053/j.gastro.2017.06.012
- 15. van der Meer AJ, Feld JJ, Hofer H, et al. Risk of cirrhosis-related complications in patients with advanced fibrosis following hepatitis C virus eradication. J Hepatol. 2017;66(3):485–493. doi:10.1016/j.jhep.2016.10.017
- Mendizabal M, Pinero F, Ridruejo E, et al. Disease progression in patients with hepatitis C virus infection treated with direct-acting antiviral agents. *Clin Gastroenterol Hepatol.* 2020;18(11):2554–2563 e3. doi:10.1016/j.cgh.2020.02.044
- 17. Tahata Y, Hikita H, Mochida S, et al. Liver-related events after direct-acting antiviral therapy in patients with hepatitis C virus-associated cirrhosis. *J Gastroenterol*. 2022;57(2):120–132. doi:10.1007/s00535-021-01845-5
- Butt AA, Yan P. Natural history of hepatitis C virus infection in a large national seroconversion cohort in the direct-acting antiviral agent era: results from ERCHIVES. J Viral Hepat. 2021;28(6):916–924. doi:10.1111/jvh.13507
- 19. Tacke F, Boeker KHW, Klinker H, et al. Baseline risk factors determine lack of biochemical response after SVR in chronic hepatitis C patients treated with DAAs. *Liver Int.* 2020;40(3):539–548. doi:10.1111/liv.14186
- 20. Ghany MG, Morgan TR; panel AlhCg. Hepatitis C guidance 2019 update: American Association for the Study of Liver Diseases–Infectious Diseases Society of America recommendations for testing, managing, and treating hepatitis C virus infection. *Hepatology*. 2020;71(2):686–721.
- 21. Liver EAfTSoT. EASL recommendations on treatment of hepatitis C 2018. J Hepatol. 2018;69(2):461-511.
- 22. Lu M, Li J, Zhang T, et al. Serum biomarkers indicate long-term reduction in liver fibrosis in patients with sustained virological response to treatment for HCV infection. *Clin Gastroenterol Hepatol.* 2016;14(7):1044–1055.e3. doi:10.1016/j.cgh.2016.01.009
- Ioannou GN, Beste LA, Green PK, et al. Increased risk for hepatocellular carcinoma persists up to 10 years After HCV eradication in patients with baseline cirrhosis or high FIB-4 scores. *Gastroenterology*. 2019;157(5):1264–1278 e4. doi:10.1053/j.gastro.2019.07.033
- Thandassery RB, Kaabi SA, Soofi ME, Tharian B, Singh R. Noninvasive serum models to predict significant liver related events in chronic hepatitis C. *Hepatol Int.* 2017;11(4):401–408. doi:10.1007/s12072-017-9800-7
- Marrero JA, Kulik LM, Sirlin CB, et al. Diagnosis, staging, and management of hepatocellular carcinoma: 2018 practice guidance by the American association for the study of liver diseases. *Hepatology*. 2018;68(2):723–750. doi:10.1002/hep.29913
- 26. Sterling RK, Lissen E, Clumeck N, et al. Development of a simple noninvasive index to predict significant fibrosis in patients with HIV/HCV coinfection. *Hepatology*. 2006;43(6):1317–1325.
- Kanwal F, Kramer JR, Asch SM, Cao Y, Li L, El-Serag HB. Long-term risk of hepatocellular carcinoma in HCV patients treated with direct acting antiviral agents. *Hepatology*. 2020;71(1):44–55. doi:10.1002/hep.30823
- R Core Team. R: a language and environment for statistical computing. R Foundation for statistical computing, Vienna, Austria; 2022. Available from: https://www.R-project.org/. Accessed December 20, 2022.
- Singal AG, Rich NE, Mehta N, et al. Direct-acting antiviral therapy for hepatitis C virus infection is associated with increased survival in patients with a history of hepatocellular carcinoma. *Gastroenterology*. 2019;157(5). doi:10.1053/j.gastro.2019.07.040
- Ioannou GN, Green PK, Berry K. HCV eradication induced by direct-acting antiviral agents reduces the risk of hepatocellular carcinoma. J Hepatol. 2017. doi:10.1016/j.jhep.2017.08.030
- Wang S, Liu Y, Feng Y, et al. A review on curability of cancers: more efforts for novel therapeutic options are needed. *Cancers*. 2019;11(11). doi:10.3390/cancers11111782
- Boursier J, Brochard C, Bertrais S, et al. Combination of blood tests for significant fibrosis and cirrhosis improves the assessment of liver-prognosis in chronic hepatitis C. Aliment Pharmacol Ther. 2014;40(2):178–188. doi:10.1111/apt.12813
- 33. Toyoda H, Tada T, Yasuda S, Mizuno K, Ito T, Kumada T. Dynamic evaluation of liver fibrosis to assess the risk of hepatocellular carcinoma in patients with chronic hepatitis C who achieved sustained virologic response. *Clin Infect Dis.* 2020;70(6):1208–1214. doi:10.1093/cid/ciz359
- Na SK, Song B-C. Development and surveillance of hepatocellular carcinoma in patients with sustained virologic response after antiviral therapy for chronic hepatitis C. Clin Mol Hepatol. 2019;25(3):234–244. doi:10.3350/cmh.2018.0108
- 35. Degasperi E, D'Ambrosio R, Iavarone M, et al. Factors associated with increased risk of de novo or recurrent hepatocellular carcinoma in patients with cirrhosis treated with direct-acting antivirals for HCV infection. *Clin Gastroenterol Hepatol*. 2019;17(6):1183–1191 e7. doi:10.1016/j.cgh.2018.10.038

Journal of Hepatocellular Carcinoma



#### Publish your work in this journal

The Journal of Hepatocellular Carcinoma is an international, peer-reviewed, open access journal that offers a platform for the dissemination and study of clinical, translational and basic research findings in this rapidly developing field. Development in areas including, but not limited to, epidemiology, vaccination, hepatitis therapy, pathology and molecular tumor classification and prognostication are all considered for publication. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/journal-of-hepatocellular-carcinoma-journal

If y in DovePress

1443